Author’s response to reviews

Title: Endoscopic Examination of Labial Fusion in a Postmenopausal Woman: A Case Report

Authors:
Yusaku Kumagai (yusakumagai@med.tohoku.ac.jp)
Masafumi Toyoshima (m-toyo@med.tohoku.ac.jp)
Kei Kudo (kei.kudo.b3@tohoku.ac.jp)
Minoru Ohsawa (minor.oh38@med.tohoku.ac.jp)
Hitoshi Niikura (niikura@med.tohoku.ac.jp)
Nobuo Yaegashi (yaegashi@med.tohoku.ac.jp)

Version: 1 Date: 11 Dec 2017

Author’s response to reviews:

Reviewer #2:

4. Does the article report the following information? Where information is missing, please specify.

Comments: some consideration is needed about obstetric antecedents, since labial fusion is related to genitalia trauma.

Thank you for your important comments. The patient has no history of trauma at birth. We stated this in the P5 Line58-59.

9. Additional comments for the author(s)?

Few reports can be found in the web relative to labial fusion. Most of them from younger patients and eventually in elderly people. This later group show fusion as a result of estrogen deprivation, genitalia trauma or inflammatory diseases. Since no history of the latest two is refered in the paper, only estrogen reduction could be involved in this case. Nevertheless, it would have
been of great contribution to explore the conditions in which her deliveries concluded because no mention is made respect to pelvic floor trauma or infectious diseases in puerperium.

We agree the reviewer’s comments and stated the patient’s medical history concerning pelvic trauma or infectious disease in the external genitalia in P5 Line 59-60.